Author's response to reviews

Title: Comparison of airway measurements during influenza-induced tachypnea in infant and adult cotton rats

Authors:

Elman L Trias (elmantrias@gmail.com)
Arash Hassantoufighi (Arash.Hassantoufighi@fda.hhs.gov)
Gregory A Prince (gprince@erols.com)
Maryna C Eichelberger (Maryna.Eichelberger@fda.hhs.gov)

Version: 6 Date: 28 May 2009

Author's response to reviews:

The following changes were made:
Affiliations - Each affiliation is now written in full.
Tables - I have used MS Word auto-format template: simple 1. Please do not include vertical lines in your tables.
Figure numbers were removed from the figure. Figures were cropped.
I corrected reference numbers in the text.
I removed 'Address, E-mail' subheadings from the title page.
I deleted 'Trias et al' from the head of every page.
I checked for and correct a few typographical errors